Author's response to reviews

Title: Prostate cancer: net survival and cause-specific survival rates after multiple imputation.

Authors:

Adeline Morisot (adeline.morisot@gmail.com)
Faïza Bessaoud (faiza.bessaoud@orange.fr)
Paul Landais (paul.landais@chu-nimes.fr)
Xavier Rébillard (xavier.rebillard@wanadoo.fr)
Brigitte Trétarre (registre-tumeur@wanadoo.fr)
Jean-Pierre Daurès (jeanpierre.daures@sfr.fr)

Version: 3
Date: 10 June 2015

Author's response to reviews: see over
Dear Mrs. Pura,

Thank you very much for accepting this paper for publication. The following lines were added at the end of the Methods section, in order to answer your inquiries regarding data availability:

Data used in this study are publicly available and approved by two ethical committees for studies using human subjects (National Data Processing Consultative Committee for Medical Research-CCTIRS- and by the Commission Nationale Informatique et Libertés-CNIL) which provided approval to access at population-based cancer, ERSPC and RNIPP data in Hérault and advocates that all medical information are confidential and anonymous (declaration n°900075).

Moreover we replaced "Répertoire National d’Identification des Personnes Physiques” by ”National Directory for the Identification of Natural Persons” and ”Centre d’Epidémiologie sur les causes médicales de décès” by ”French epidemiology centre of the medical causes of death”.

Sincerely yours,
Adeline Morisot